

Received February 3, 2020, accepted February 16, 2020, date of publication February 21, 2020, date of current version March 3, 2020. Digital Object Identifier 10.1109/ACCESS.2020.2975663

Flexibility Services Management Under Uncertainties for Power Distribution Systems: Stochastic Scheduling and Predictive Real-Time Dispatch

VASILEIOS A. EVANGELOPOULOS[®], IASON I. AVRAMIDIS[®], (Student Member, IEEE), AND PAVLOS S. GEORGILAKIS[®], (Senior Member, IEEE)

School of Electrical and Computer Engineering, National Technical University of Athens (NTUA), Athens 157 80, Greece

Corresponding author: Pavlos S. Georgilakis (pgeorg@power.ece.ntua.gr)

This research was co-financed by the European Union and Greek national funds through the Operational Program Competitiveness, Entrepreneurship and Innovation, under the call RESEARCH – CREATE – INNOVATE (project code: T1EDK-00450).

ABSTRACT Over the last years, the role of the distribution system operator (DSO) has largely expanded. This is necessitated by the increased penetration of intermittent energy resources at the distribution level, as well as the new, more complex interactions with the transmission system operator (TSO). As such, to properly manage its system and to have an effective joint cooperation with the TSO, the DSO is required to procure and carefully manage flexibility services from distributed energy resources (DER). This paper introduces a thorough framework on optimal operational planning (day-ahead scheduling) and operational management (real-time dispatch) of active distribution systems under uncertainties, to avoid line congestions and voltage limit violations, and efficiently balance the distribution system. A two-stage stochastic programming model based on weighted scenarios is proposed to optimize the multi-period optimal power flow day-ahead scheduling, i.e., scheduled power flows at the TSO-DSO interface and reserved DER flexibility services. Subsequently, the operational management, realized with a predictive real-time dispatch model based on a constantly updated rolling horizon, aims to efficiently activate the available flexibility services to minimize deviations from the committed schedule. Different sources of flexibility are considered, with their respective response times also taken into account at real-time dispatch. The proposed framework is applied on two distribution systems and investigates the DSO's level of risk exposure while minimizing its total cost (reservation and activation expenses). The results indicate that a less conservative approach at planning stage, despite the potential risk exposure, can lead to significant reduction in total expenses.

INDEX TERMS Battery storage systems, distributed energy resources, flexibility services management, real-time dispatch, renewable energy sources, stochastic day-ahead scheduling, stochastic programming.

NOMENCLATURE		FSP	Flexibility service providers
ACRONYMS & ABBREVIATIONS		FSU	Flexibility service upward-regulated
ANM	Active network management	HV	High voltage
BSS	Battery storage systems	LV	Low voltage
DAS	Dav-ahead scheduling	MPC	Model predictive control
DER	Distributed energy resources	MV	Medium voltage
DERA	DFR aggregators	NLP	Non-linear programming
DSO	Distribution system operator	OPF	Optimal power flow
FSD	Elevibility service downward-regulated	PV	Photovoltaics
150	The storing service downward-regulated	P-RTD	Predictive real time dispatch
The associate editor coordinating the review of this manuscript and		RES	Renewable energy sources
approving it for publication was Pierluigi Siano		RTD	Real time dispatch

Ch/Dch

SS	Substations
S-DAS	Stochastic day-ahead scheduling
TSO	Transmission system operator
T-D	TSO-DSO interface
WG	Wind generators

SETS & INDICES

Ω_N	Buses of the system, indexed by <i>i</i> and <i>j</i>
Ω_{Ni}	Buses of the system connected with bus <i>i</i>
Ω_D	Loads, indexed by d

- Ω_{Di} Loads located at bus *i*
- Ω_W RES, indexed by w
- Ω_{Wi} RES located at bus *i*
- Ω_B BSS, indexed by b
- Ω_{Bi} BSS located at bus *i*
- Ω_{AG} DERA, indexed by ag
- Ω_{AGi} DERA regulating the net load of bus *i*
- Φ HV/MV SS of the system, indexed by *ss*
- Φ_i HV/MV SS of the system located at bus *i*
- S Scenarios, indexed by s
- S_{ℓ} States of load, indexed by ℓ
- S_v States of wind speed, indexed by v
- *t* Index of time-intervals, running from 1 to *T*
- *h* Index of rolling horizon time-intervals, running from 1 to *H*

PARAMETERS

G_{ii}/B_{ii}	Conductance/susceptance of line $i - j$
S_{ii}^{\max}	Capacity of line $i - j$ [MVA]
S_{ss}^{max}	Capacity of HV/MV SS ss [MVA]
$V^{\max/\min}$	Maximum/minimum voltage magnitude
$P_{d,t}^{frc}$	Forecast of active power consumed by load <i>d</i> in time-interval <i>t</i> [MW]
$P_{d,t}^{act}$	Actual active power consumed by load d in time-interval t [MW]
$P_{w,t}^{frc}$	Forecast of available active power pro- duced by RES <i>w</i> in time-interval <i>t</i> [MW]
$P_{w,t}^{act}$	Actual available active power produced by RES <i>w</i> in time-interval <i>t</i> [MW]
E_{h}^{Cap}	Energy capacity of BSS b [MWh]
P_{h}^{Cap}	Power capability of BSS b [MW]
$SoC_b^{\max/\min}$	Maximum/minimum state of charge of BSS <i>b</i>
$\eta_{h}^{Ch/Dch}$	Charging/discharging efficiency of BSS <i>b</i>
$E_{ag}^{FSU/FSD}$	Energy capacity of FSU/FSD offered by DERA <i>ag</i> [MWh/day]
$P_{ag}^{FSU/FSD}$	Power capability of FSU/FSD of DERA <i>ag</i> [MW]
$\Delta P_{ag}^{FSU/FSD}$	Ramping limit of FSU/FSD of DERA <i>ag</i> [MW/h]
$C_{ag}^{R-FSU/FSD}$	Cost of reserved FSU/FSD of DERA <i>ag</i> [€/MW-h]
$C_{ag}^{A-FSU/FSD}$	Cost of activated FSU/FSD of DERA <i>ag</i> [€/MWh]

c_b	$b \in MWh$
C_b^R	Cost of reserved capacity of BSS b
C_w^{curt}	[€/MW-day] Cost of wind generation curtailment [€/MWh]
$ ho_{ss,t}^{Dev+/Dev-}$	Penalty for upward/downward deviation at
_	I-D in time-interval $t \in [MWh]$
π_s	Probability of load state ℓ
π_{ℓ}	Probability of wind speed state v
$\frac{d_{V}}{d_{t}}$	Duration of time-interval <i>t</i> [hour]
VARIABLES	
$V_{i,t}/\theta_{i,t}$	Voltage magnitude/angle of bus t in time- interval t
$P_{ii,t}/Q_{ii,t}$	Active/reactive power flow of line $i - j$ in
57.7 - 57.	time-interval t [MW/MVAr]
$P_{ss,t}^{DAS}$	Active power scheduled to be imported
	at 1-D, i.e., at the HV/MV SS, in time-
$P_{\rm e}/O_{\rm e}$	Active/reactive power imported at T-D in
$1 ss, t \neq Ss, t$	time-interval <i>t</i> [MW/MVAr]
$P_{ss t}^{Dev+/Dev-}$	Upward/downward deviation of active
55,1	power imported at T-D in time-interval <i>t</i> [MW]
P_{int}^{Inj}	Active power injected by RES w in time-
- <i>W</i> , <i>l</i>	interval t [MW]
$Q_{w,t}^{Inj}$	Reactive power of RES <i>w</i> in time-interval
$E_{b,t}$	Energy stored in BSS <i>b</i> at the end of time-
D. FOULFOD	interval t [MWh]
$P_{ag,t}^{R-FSU/FSD}$	Reserved FSU/FSD of DERA <i>ag</i> in time- interval t [MW]
$p^{A-FSU/FSD}$	A stivuted ESLIZED of DEDA as in time
P _{ag,t}	interval t [MW]
$O_{aa,t}^{FSU/FSD}$	Reactive power of FSU/FSD of DERA ag
∠ ug,i	in time-interval t [MVAr]
E_{h}^{R-Cap}	Reserved energy capacity of BSS b [MWh]
P_{h}^{R-Cap}	Reserved power capability of BSS <i>b</i> [MW]
P_{L}^{b}	Charging/discharging active power of BSS
• b,t	<i>b</i> in time-interval <i>t</i> [MW]

Cost of charging/discharging of RSS

I. INTRODUCTION

The ever-increasing integration of renewable DER due to the deregulation of the energy market creates more challenges for the DSO, as high levels of uncertainty are introduced into the electrical power system, especially at the distribution level. Apart from the traditional problem of voltage control in passive distribution systems, the issue of congestion management has also attracted significant attention. In adhering with the future smart grid vision [1], active distribution systems are expected to host very high shares of renewable technologies, which have been noted to cause substantial thermal issues if

left uncontrolled. It comes to no surprise that recent research works have proposed approaches to combat congestions [2]–[6]. On the other hand, DER are also excellent sources of flexibility for managing the electrical systems. The DSO can utilize DER flexibility to optimize its system's voltage profile, manage local congestions [7], and generally optimize the operation of smart distribution networks through ANM schemes [8].

However, the uncoordinated activation of DER flexibility by the DSO is likely to impact the system's balance and imply extra costs to the TSO, as explained in [7]. A prerequisite for the beneficial use of DER flexibility is the joint cooperation with the TSO, a topic that, recently, is widely discussed by many studies [9]–[11], with several coordination schemes having been proposed [12]. For the purposes of this paper, the TSO-DSO coordination is realized under the scheme of shared balancing responsibility model, according to the conceptual framework of [12]. Under this coordination scheme, the DSO organizes a local flexibility services market and is responsible for congestion management and balancing the distribution grid, while staying committed to a predefined schedule in the interconnection with the TSO.

This paper aims to propose a comprehensive framework of flexibility services management for the DSO. Within this framework, the DSO firstly carries out the DAS of the necessary DER flexibility for congestion management and balancing the local system, while optimizes an import/export hourly schedule at the T-D to timely notify it to the TSO. Subsequently, the distribution system is operated by proper RTD of the reserved DER flexibility. The overall objective is to minimize the deviations from the committed import/export schedule at the T-D with minimum flexibility services costs (reserve and activation cost), while meeting all technical limits.

In the past, the DAS was modelled as a multi-period OPF problem, assuming negligible forecast error, and thus ignoring the inherent uncertainties of RES and loads. Articles [13]–[15] are representative of such a DAS approach presenting deterministic models that anticipate the periods when network technical constraints are violated and propose volt/var control and/or up/down active power regulation for congestion management during these periods. In [13], the DAS is modelled as an active-reactive OPF that considers the four-quadrant operation capability of the BSS to maximize wind energy yield. A dynamic OPF formulation with energy storage and flexible demand in an ANM framework is proposed in [14]. An integer genetic algorithm is employed to optimally solve the DAS problem and minimize operational costs in [15].

In recent years, research has focused on new models that integrate in DAS the uncertainties of RES and load. The authors of [16] propose a deterministic, dynamic OPF model that considers the operational uncertainties of wind generation and demand during the planning stage. This model schedules the import/export at the T-D and commits reserves for the worst-case uncertainty scenario. Such an approach leads to more expensive reserves than an approach that would consider more scenarios of uncertainty along with their probability of occurrence. In [17], a two-stage stochastic multi-period OPF is used to schedule the generation of the dispatchable units given the available demand response flexibility, while considering the wind power uncertainty under multiple weighted scenarios. The authors of [18] introduce a two-stage robust OPF model for DSOs willing to utilize flexibility services from local DER to solve congestion/voltage problems considering the worst-case scenario of uncertainty. Even though the approach of [18] concludes in more reliable solutions, it also leads to more expensive reserves and with total operational costs similar to those of a deterministic scheduling with OPF.

The DAS is by itself insufficient to properly manage the distribution system, and thus, RTD is additionally employed to achieve improved operation. Several dispatch techniques have been proposed, from single-period OPF models to more advanced techniques of predictive control based on multiperiod OPF models with a rolling horizon. RTD with predictive control is based on the principles of MPC, a model that predicts the future evolution of a process to optimize a set of control variables over a constantly updated rolling horizon.

Several studies have employed the MPC approach to manage the operation of power systems [19]-[23]. In [19], a MPC algorithm is proposed to solve the economic dispatch problem in the presence of RES in electrical systems, while the authors of [20] propose a centralized MPC strategy to maximize the wind energy yield in an isolated electrical system by compensating for wind and load variability. Studies [19]-[20] both highlight the potential of MPC in electrical systems, but overlook important technical aspects, such as voltage and power flow limits, without having formulated the power flow equations. A combined day-ahead and intra-day hierarchical optimization is introduced in the works of [21]-[22] to maximize RES exploitation, while minimizing variations between DAS and RTD. A MPC-based dispatch approach addresses the uncertainties of electric vehicle charging patterns in [23].

However, none of the proposed models has considered the response time of DER offering flexibility services, which is proposed as a new flexibility index by [24]. The response time is also presented by [25] as a physical requirement of flexibility that should be taken into account by the operators. Both [24] and [25], highlight the significance of integrating the response time in the modelling of flexibility services. To the authors' knowledge, this is the first paper to take into account the response time of flexibility services in the MPC algorithm to solve the RTD problem, with respect to DSO utilizing flexibility services from DER.

The contribution of this paper is threefold:

a) It proposes a novel framework for managing flexibility services that optimizes both the DAS of the reserved flexibility as well as their activation during RTD procedure. The proposed framework additionally investigates the DSO's level of risk exposure that minimizes the total expenditures of flexibility services (DAS and RTD costs).

b) It proposes a novel stochastic DAS approach based on a two-stage stochastic programming formulation. The proposed S-DAS schedules the power flows at the T-D and minimizes the reserved flexibility services by minimizing the cost of their expected activation.

c) A predictive control strategy is employed for the RTD procedure. The proposed P-RTD not only considers the modelling of the ramping limits and the dynamic operation of DER, but also proposes a model for the efficient handling of the different response times of system's flexibility sources.

The remainder of this paper is structured as follows: Section II presents the proposed framework for flexibility services management. Section III formulates the proposed S-DAS. Section IV describes the proposed P-RTD optimization procedure. Section V presents the examined case studies and analyses the obtained results. Section VI concludes the paper and proposes future research work.

II. PROPOSED FRAMEWORK FOR FLEXIBILITY SERVICES MANAGEMENT

This paper proposes a framework for optimal operational planning (DAS stage) and optimal operational management (RTD stage) of active distribution systems by means of congestion management and system balancing, while maximizing the active power injection from RES. With this approach, the DSO exploits the DER flexibility services in the most effective way in order to stay committed to the hourly import/export schedule at the T-D with minimum deviations. The expected value of these deviations is minimized at the DAS stage by anticipating the needed flexibility services, whereas the actual value of these deviations is minimized during RTD by activating the needed portions of flexibility services from those reserved during DAS.

A. MAIN ASSUMPTIONS

For the sake of clarity, the main assumptions adopted in this paper are summarized as follows:

a) The FSP included in this study are: i) DERA (aggregators of flexible loads and DER operating at LV), ii) BSS (operating at MV), and iii) RES (operating at MV). Their models are described in Section III.

b) The proposed framework is applied under normal operating conditions, where the scenarios considered are solely related to the intermittent nature of RES and load. Extreme conditions, such as WG tripping, which require contingency reserves, are not within the scope of this paper.

c) Wind power generation depends, among other factors, on various parameters of weather conditions. In this paper, for simplicity, the authors have considered that wind power generation depends only on wind speed and thus, is calculated based on the well-known wind turbine S-curve $P_w(v)$ [13]. Additionally, without loss of generality, wind speed profile is considered the same across all buses of the distribution system.

d) It is supposed that an online local flexibility market platform has been developed. This platform is intended to facilitate the trading between DSO and the FSP.

e) The FSP are able to submit their bids in the local flexibility market platform until 10:00 am. After the local market is cleared, the DSO timely notifies the scheduled power flows at the T-D to the TSO, not later than 12.00 at noon. The TSO, in turn, clears the ancillary services day-ahead market under its responsibility, having the predefined schedules at the T-D.

B. FLEXIBILITY SERVICE PROVIDERS

A flexibility service, as defined in [26], is a power regulation (upward or downward) sustained at a given time-period for a predefined duration from a specific location within the network. The flexibility service, as a product, is technically characterized by four indices: a) power capability, b) energy capacity, c) ramping limit and d) response time [24].

In what follows, flexibility services are split in two main categories according to the direction of active power regulation. Thus, the term FSD refers to demand reduction (or generation increase) and vice versa for the term FSU.

1) DER AGGREGATORS

A DERA is an intermediary entity that aggregates the demand of end-customers' flexible loads and the generation of residential DERs (e.g., roof-top photovoltaics, other small-scale RES) to provide both FSD and FSU to the DSO upon request. The DERA provides its services at the MV level of the distribution system. The DSO, in turn, offers proper remuneration, according to the DERA's bid (energy-price pair), depending on the reserved and the activated volumes of FSD and/or FSU. Therefore, the DERA is remunerated at a "commitment price" (in €/MW-h) for keeping the needed FSD and FSU reserves, and at an "activation price" (in €/MWh) for the finally delivered FSU and FSD.

FSD offered by DERA can be achieved by reducing customers' aggregated consumption, or by increasing the power injected by local DER (e.g., discharging of the scheduled stored energy for backup). FSU, in accordance, is derived by increasing customers' aggregated consumption (e.g., through price signals or demand response programs) or by reducing residential RES aggregated production (e.g., generation curtailment). The way in which the DERA achieve FSU and FSD is not the focus of this paper.

2) BATTERY STORAGE SYSTEMS

BSS can inherently provide bi-directional flexibility services, both FSD and FSU, when discharging and charging, respectively. It is assumed that BSS are privately-owned, and an agreement is in place stating that the DSO may reserve partial of full capacity of the BSS at the day-ahead stage. The BSS are then dispatched by the DSO to "absorb" any power mismatches from the day-ahead committed schedule. The portion of BSS's capacity that is scheduled to be utilized is determined during DAS. In case the capacity of the BSS is partially reserved, it is assumed that its owner uses the remaining capacity in a way that does not affect the corresponding bus's net load profile. Here, again a twofold remuneration scheme is in place for the services provided by BSS owners. The BSS owner is remunerated by the DSO at a "commitment price" (in \in /MW-day) for its capacity availability and at an "activation price" (in \notin /MWh) when charging/discharging active power is activated.

3) RENEWABLE ENERGY SOURCES

Privately-owned large-scale RES connected at MV of the distribution system are subject to curtailment providing a service of FSU. RES curtailment is the DSO's last resort, as it is usually activated in case of forecasted congestion or when energy mismatches between DAS and RTD are more expensive than RES tariffs. The conditions under which a DSO can force curtailment are determined in a power purchase agreement contract between the DSO and the RES owner. Curtailment policies, as well as the compensation paid to the RES owners differ per country [27]. In this paper, the cost of RES curtailment is set equal to a fixed price (in \in/MWh), supposing a feed-in-tariff contract.

Large-scale RES should also be able to provide reactive power support according to grid code. The reactive power requirements, as specified in different grid codes, are summarized in [28]. For the purposes of this study, the reactive power capability ranges $\pm 33\%$ of the active power injected by RES, i.e., power factor ranges from 0.95 leading to 0.95 lagging. This is a producer's obligation and thus the DSO uses reactive power compensation free of remuneration.

C. STRUCTURE OF THE PROPOSED FRAMEWORK

Fig. 1 depicts the proposed framework that structures the solution of the problem in two stages.

1) DAY-AHEAD SCHEDULING (1ST STAGE)

Formulated as a multi-period OPF, in the general approach, DAS comprises the decisions to be made one day in advance given the day-ahead forecasting. In this paper, a two-stage stochastic programming model for the multi-period OPF, hereafter called S-DAS, is proposed to optimize the first stage variables, while ensuring feasible control actions throughout the second stage. To simulate the uncertain realization of the second stage, multiple scenarios are considered, based on the day-ahead forecast, weighted according to their probability of occurrence.

The first stage variables, pertaining to *here-and-now* decisions that ensure feasible solution across all scenarios considered, are the following:

- 1) scheduled import at T-D, i.e., the HV/MV SS,
- 2) reserved FSD and FSU by DERA,
- 3) reserved BSS power capability.

The vector of *here-and-now* control variables in time-interval *t* is defined as follows:

$$\mathbf{u}_{t}^{DAS} = \left\{ P_{ss,t}^{DAS}, P_{ag,t}^{R-FSU}, P_{ag,t}^{R-FSD}, P_{b}^{R-Cap} \right\}$$



FIGURE 1. Proposed framework for flexibility services management.

The second stage variables, referring to *wait-and-see* decisions potential to be realized during RTD, comprise the control actions activated by the DSO:

- 1) DERA FSD and FSU activation,
- 2) BSS active power charging/discharging,
- 3) RES active power curtailment,
- 4) RES reactive power compensation,

and the state variables of the system during RTD:

- 5) voltage magnitude and angle per bus,
- 6) active and reactive power flow per line,
- 7) active and reactive power imported at the T-D,
- 8) reactive power of activated FSD and FSU,
- 9) BSS stored energy at the end of each time-interval.

The vectors of the second stage variables (control and state variables) in time-interval t under scenario s that are used in the S-DAS are defined as follows:

$$\mathbf{u}_{t,s}^{RTD} = \left\{ P_{ag,t,s}^{A-FSU}, P_{ag,t,s}^{A-FSD}, P_{b,t,s}^{Ch}, P_{b,t,s}^{Dch}, P_{w,t,s}^{Inj}, Q_{w,t,s}^{Inj} \right\} \\ \mathbf{x}_{t,s}^{RTD} = \left\{ \begin{array}{l} V_{i,t,s}, \theta_{i,t,s}, P_{ij,t,s}, Q_{ij,t,s}, P_{ss,t,s}^{Dev+}, P_{ss,t,s}^{Dev-}, \\ Q_{ss,t,s}, Q_{ag,t,s}^{FSU}, Q_{ag,t,s}^{FSD}, E_{b,t,s} \end{array} \right\}$$

2) REAL-TIME DISPATCH (2ND STAGE)

The RTD stage is the validation process of the DAS. In this stage, the DSO transitions to the RTD, given the reserves and the schedule of DAS. In general, RTD is an iterative procedure that determines the control actions to be implemented at each time-interval of the operation based on the decisions made in the DAS stage and given the updated state of the system. In this paper, a P-RTD is proposed, formulated as multi-period OPF to take advantage of a look-ahead rolling

horizon, based on the principles of MPC. The modelling of response time of the available flexibility services is also introduced. The P-RTD utilizes the constantly refreshed shortterm forecasting to determine the real-time control actions, anticipating optimal future control actions of the rolling horizon.

The P-RTD, thoroughly analysed in Section IV, optimizes the vectors of the second stage variables $\mathbf{u}_{t+h|t}^{P-RTD}$ and $\mathbf{x}_{t+h|t}^{P-RTD}$ at the *h*-th time-interval of the *t*-th rolling horizon, given the system's current state, i.e., for h = 0 of the previous rolling horizon, $\mathbf{u}_{t|t-1}^{P-RTD}$ and $\mathbf{x}_{t|t-1}^{P-RTD}$.

$$\mathbf{u}_{t+h|t}^{P-RTD} = \begin{cases} P_{ag,t+h|t}^{A-FSU}, P_{ag,t+h|t}^{A-FSD}, P_{b,t+h|t}^{Ch}, \\ P_{b,t+h|t}^{Dch}, P_{w,t+h|t}^{hj}, Q_{w,t+h|t}^{hj} \end{cases}$$
$$\mathbf{x}_{t+h|t}^{P-RTD} = \begin{cases} V_{i,t+h|t}, \theta_{i,t+h|t}, P_{ij,t+h|t}, Q_{ij,t+h|t}, Q_{ss,t+h|t}, \\ P_{ss,t+h|t}^{Dev+}, P_{ss,t+h|t}^{Dev-}, Q_{ag,t+h|t}^{FSU}, Q_{ag,t+h|t}^{FSD}, E_{b,t+h|t} \end{cases}$$

III. S-DAS FORMULATION

In this section, firstly the uncertainties of RES and load are modelled by creating different scenarios. Subsequently, the models of the FSP, the interaction with the TSO and the network constraints are described. The objective function and the overall S-DAS formulation are also given.

A. DAY-AHEAD FORECASTING UNCERTAINTIES MODELLING

This subsection explains the process of scenario creation for handling the uncertainties of the day-ahead point forecasts of load and RES generation. These scenarios represent scenarios of the distribution system's net load, since they are derived from combinations of different uncertain states of load and RES generation. Their values are used as input parameters for the S-DAS.

1) PROBABILISTIC FORECASTS WITH PREDICTION INTERVALS

In decision-making, probabilistic forecasting provides a better view of the potential future outcomes than deterministic forecasting. The term *deterministic forecasting* refers to methods for producing point forecasts, which represent the conditional expectations of the future outcomes without considering future errors, whereas the term *probabilistic forecasting* is used when prediction error and its probability density function are considered.

A prediction interval is a type of probabilistic forecast directly linked to past errors due to point forecasts, which do not consider the error distributions [29]. According to [30], a prediction interval with a coverage rate α % is defined as a range of potential values for the future observations y_{t+h} , computed based on past observations, such that the future observations y_{t+h} will fall in this interval with probability α , where $\alpha \in [0, 1]$. For example, for a prediction interval with a coverage rate of 90%, there is 90% probability for the future observations to belong in this prediction interval [30]. Prediction intervals are usually based on the root mean square error of the deterministic forecasting model, which provides an estimate of the standard deviation of the forecast error [31]. In this paper, the upper and lower bounds of the prediction interval are constructed by (1) assuming that the forecast error of the h – th next time-interval is normally distributed with zero mean and standard deviation σ_h , i.e., $\varepsilon_h \sim N(0, \sigma_h^2)$ [32].

$$\hat{y}_{t+h} \pm z\sigma_h \tag{1}$$

where \hat{y}_{t+h} denotes the point forecast issued at time-interval *t* for the *h*-th next time-interval and *z* is the critical value of the standard normal distribution that gives the coverage rate of the prediction interval. For example, *z* = 3 gives a prediction interval with a coverage rate of 99.7% [31]. Do note that the standard deviation of the forecast error depends on the lead time of the prediction.

In what follows, the probabilistic forecasts of the future outcomes are realized based on a prediction interval with a certain coverage rate. This prediction interval, in turn, is used to create multiple representative states of the future outcomes, each one with its probability of occurrence. These states are created by dividing the selected part of the normal distribution of the forecast error (i.e., the area within the prediction interval selected) into distinct intervals, according to the methodology followed in [33] and [34].

Wind speed and load forecast errors are both assumed to follow the normal distribution, a common approach employed by many researchers to model such forecast errors [34]–[36]. Modifications on the probability distribution function of the forecast errors could be made accordingly to better approximate the reality; however, this is out of the scope of this paper. Nonetheless, the forecast error modelling (for both wind speed and load) that is detailed hereafter is generic; it is applicable for any distribution function desired, should one wish to obtain even more representative results.

2) WIND SPEED FORECASTING UNCERTAINTIES

The uncertainties of wind speed deterministic forecasting are strongly related to the forecast error of the numerical weather prediction tools. This error is assumed to be normally distributed with zero mean and standard deviation σ_w [35].

To construct multiple states of wind speed, firstly, the normal distribution of the wind speed forecast error is divided into distinct intervals. An example of nine intervals is presented in Table 1. The corresponding value of wind speed forecast error for each state (ε_v^w) is determined taking into account the worst-case per state in accordance with Table 1. For example, in state 1, where $\varepsilon_v^{lo} = -3\sigma_w$ and $\varepsilon_v^{up} = -2\sigma_w$, it implies that $\varepsilon_v^w = -3\sigma_w$, i.e., ε_v^w takes the worst case value between ε_v^{lo} and ε_v^{up} . Similarly, in state 8, where $\varepsilon_v^{lo} =$ $+1.5\sigma_w$ and $\varepsilon_v^{up} = +2\sigma_w$, it implies that $\varepsilon_v^w = +2\sigma_w$. The probability of each state is calculated with (2a). The sum of all probabilities of the selected states is equal to the coverage rate a_V of the considered prediction interval, as (2b) shows. For

 TABLE 1. States of wind speed forecast error for prediction interval with coverage rate of 99.7%.

~	Lower limit	Upper limit	Value	Probability
State	(\mathcal{E}_{v}^{lo})	$\left(\left. \mathcal{E}_{v}^{up} \right. ight)$	(\mathcal{E}_{v}^{w})	(π_v)
1	$-3\sigma_w$	$-2\sigma_w$	$-3\sigma_w$	0.021
2	$-2\sigma_w$	$-1.5\sigma_w$	$-2\sigma_w$	0.044
3	$-1.5\sigma_{w}$	$-1\sigma_w$	$-1.5\sigma_w$	0.092
4	$-1\sigma_w$	$-0.5\sigma_w$	$-1\sigma_w$	0.150
5	$-0.5\sigma_{w}$	+0.5 σ_{w}	0	0.383
6	+0.5 $\sigma_{_w}$	$+1\sigma_{w}$	$+1\sigma_{w}$	0.150
7	$+1\sigma_{w}$	$+1.5\sigma_{w}$	+1.5 σ_w	0.092
8	$+1.5\sigma_{w}$	$+2\sigma_w$	$+2\sigma_w$	0.044
9	$+2\sigma_w$	$+3\sigma_w$	$+3\sigma_w$	0.021
			Sum (a_v)	0.997

example, the nine states of Table 1 have a prediction interval with a coverage rate (a_V) of 99.7%.

$$\pi_{\nu} = \Pr(\varepsilon_{\nu}^{lo} \le x \le \varepsilon_{\nu}^{up}) = \frac{1}{\sigma_{w}\sqrt{2\pi}} \int_{\varepsilon_{\nu}^{lo}}^{\varepsilon_{\nu}^{up}} e^{-\frac{(x-\mu_{w})^{2}}{2\sigma_{w}^{2}}} dx, \ \forall \nu \ \text{(2a)}$$
$$a_{V} = \sum_{\nu \in S_{\nu}} \pi_{\nu} \tag{2b}$$

Following this process, a set of states for the wind speed forecast error is created, each state with its value and its probability of occurrence: $\{\varepsilon_v^w, \pi_v\}$.

3) LOAD FORECASTING UNCERTAINTIES

The DSO should also predict the day-ahead demand at each of the system's buses. Assuming that the load forecast error follows the normal distribution, the same approach with wind speed probabilistic forecasting is applied. Likewise, the example of Table 1 is applied for the states of load forecast error. The probability of each state is calculated with (3a), while the sum of all probabilities of the selected states is equal to the coverage rate a_L of the considered prediction interval, as shown by (3b). A set of states for the load forecast error is created, each state with its value and its probability of occurrence: $\{\varepsilon_\ell^d, \pi_\ell\}$.

$$\pi_{\ell} = \Pr(\varepsilon_{\ell}^{lo} \le x \le \varepsilon_{\ell}^{up}) = \frac{1}{\sigma_{\ell}\sqrt{2\pi}} \int_{\varepsilon_{\ell}^{lo}}^{\varepsilon_{\ell}^{up}} e^{-\frac{(x-\mu_{\ell})^2}{2\sigma_{\ell}^2}} dx, \ \forall \ell \ (3a)$$
$$a_{L} = \sum_{\ell \in S_{\ell}} \pi_{\ell} \tag{3b}$$

4) SCENARIOS CREATION

For the creation of scenarios, each state of wind speed forecast error is combined with each state of load forecast error, i.e., $\{\varepsilon_s^w, \varepsilon_s^d\} = \{\varepsilon_v^w, \varepsilon_l^d\}_s$, using a scenario tree model as in [37]. Assuming that the individual states of wind speed and load forecast error are independent, the probability of occurrence of scenario *s* is calculated according to (4a). The load forecasting curve under scenario *s* is calculated by (4b) based on the day-ahead point forecast $\hat{P}_{d,t}^{frc}$. The wind power forecasting curve under scenario *s* is given by (4c), as a function of the wind speed day-ahead point forecast \hat{V}_t^{frc} by using the S-curve transformation of the wind turbine of WG *w*, i.e., $P_w(v)$. Equation (4d) gives the sum of all probabilities of the created scenarios. The value of β represents the probability the upcoming observations of load and wind speed to be covered by the created scenarios. Hence, β is the coverage rate of the selected scenarios.

$$\pi_s = \pi_l \cdot \pi_v \qquad \qquad \forall s \qquad (4a)$$

$$P_{d,t,s}^{frc} = \hat{P}_{d,t}^{frc} + \varepsilon_s^d \qquad \qquad \forall d, t, s \qquad (4b)$$

$$P_{w,t,s}^{frc} = P_w \left(\hat{v}_t^{frc} + \varepsilon_s^w \right) \qquad \forall w, t, s \qquad (4c)$$

$$\beta = a_L \cdot a_V = \sum_{s \in S} \pi_s \tag{4d}$$

B. FLEXIBILITY SERVICE PROVIDERS MODEL

1) DER AGGREGATORS

The participation of DERA in the day-ahead flexibility market is described by (5a)–(5e). Equations (5a) and (5b) denote the maximum energy of FSU and FSD the DSO can reserve over a day. The power capability of FSU and FSD are formulated by (5c) and (5d), accordingly. Equation (5e) gives the cost of reserves that should be remunerated.

$$\sum_{t=1}^{T} P_{ag,t}^{R-FSU} \le E_{ag}^{FSU} \qquad \forall ag \qquad (5a)$$

$$\sum_{t=1}^{I} P_{ag,t}^{R-FSD} \le E_{ag}^{FSD} \qquad \forall ag \qquad (5b)$$

$$P_{ag,t}^{R-FSU} \le P_{ag}^{FSU} \qquad \forall ag, t \quad (5c)$$

$$\begin{aligned} P_{ag,t}^{A} & \stackrel{\text{def}}{=} SD & \forall ag, t \quad (5d) \\ F^{DAS} - C^{R-FSU} P^{R-FSU} + C^{R-FSD} P^{R-FSD} & \forall ag, t \quad (5e) \end{aligned}$$

 $F_{ag,t}^{ag} = C_{ag}^{A} + S^{ag} + C_{ag}^{A} + S^{b} + C_{ag}^{A} + S^{b} + C_{ag}^{A} + S^{b} + C_{ag}^{A} + S^{b} + S^$

services provided by DERA. Depending on the hourly availability "agreed upon" during the DAS, the DSO can proceed to FSU or FSD activation during RTD, when needed, up to the reserved volume according to (6a)–(6b). The reactive power regulated by DERA, i.e., $Q_{ag,t,s}^{FSD}$ or $Q_{ag,t,s}^{FSU}$ is considered proportional to the FSD or FSU activation during RTD according to the power factor of the system's bus. Ramping limits are also considered in this stage and are modelled by (6c)–(6d). Equation (6e) gives the activation cost that should be remunerated to the DERA.

$$P_{a_{o}t}^{A-FSU} \le P_{a_{o}t}^{R-FSU} \qquad \forall ag, t, s \quad (6a)$$

$$P_{ag,t,s}^{A-FSD} \le P_{ag,t}^{R-FSD} \qquad \forall ag, t, s \quad (6b)$$

$$-d_t \Delta P_{ag}^{FSU} \le P_{ag,t,s}^{A-FSU} - P_{ag,t-1,s}^{A-FSU} \le d_t \Delta P_{ag}^{FSU} \quad \forall ag, t, s \quad (6c)$$

$$-d_t \Delta P_{ag}^{FSD} \le P_{ag,t,s}^{A-FSD} - P_{ag,t-1,s}^{A-FSD} \le d_t \Delta P_{ag}^{FSD} \quad \forall ag, t, s \quad (6d)$$
$$F_{ag,t,s}^{RTD} = C_{ag}^{A-FSU} P_{ag,t,s}^{A-FSU} + C_{ag}^{A-FSD} P_{ag,t,s}^{A-FSD} \quad \forall ag, t, s \quad (6e)$$

2) BATTERY STORAGE SYSTEMS

Equations (7a)–(7d) model the reserved BSS capacity. The DSO can fully or partially reserve the BSS power capability and the BSS energy capacity, as given by (7a) and (7b), respectively. Equation (7c) ensures that the proportion of the reserved power capability is always equal to that of the reserved energy capacity. Equation (7d) gives the cost of BSS availability calculated during the DAS.

$$P_b^{R-Cap} \le P_b^{Cap} \qquad \forall b \qquad (7a)$$

$$E_b^{R-Cap} \le E_b^{Cap} \qquad \qquad \forall b \qquad (7b)$$

$$P_b^{R-Cap}/P_b^{Cap} = E_b^{R-Cap}/E_b^{Cap} \quad \forall b$$
(7c)

$$F_b^{DAS} = C_b^R P_b^{R-Cap} \qquad \forall b \qquad (7d)$$

The operation of BSS under all scenarios is modelled by (8a)–(8f). The BSS energy balance at each time-interval should satisfy the dynamic mathematical formulas (8a)–(8b). The upper limit of power that can be absorbed by or injected to the system is described by (8c)–(8d). Equation (8e) models the upper and lower technical limits of the BSS. Equation (8f) gives the activation cost of BSS charging/discharging.

$$E_{b,t,s} = E_{b,t-1,s} + d_t \left(\eta_b^{Ch} P_{b,t,s}^{Ch} - P_{b,t,s}^{Dch} / \eta_b^{Dch} \right) \\ \forall b, s, t \neq 1 \quad (8a)$$
$$E_{b,t,s} = E_{b,0} + d_t \left(\eta_b^{Ch} P_{b,t,s}^{Ch} - P_{b,t,s}^{Ch} / \eta_b^{Dch} \right)$$

$$\forall b, s, t = 1 \quad (8b)$$

$$\forall b, t, s \quad (8c)$$

 $P_{b,t,s}^{Ch} \leq P_b^{R-Cap} \qquad \qquad \forall b, t, s \qquad (8c)$ $P_{b,t,s}^{DCh} \leq P_b^{R-Cap} \qquad \qquad \forall b, t, s \qquad (8d)$

$$SoC_b^{\min} \le E_{b,t,s} / E_b^{R-Cap} \le SoC_b^{\max} \quad \forall b, t, s$$

$$F_{b,t,s}^{RTD} = C_b^{Ch} P_{b,t,s}^{Ch} + C_b^{Dch} P_{b,t,s}^{Dch} \quad \forall b, t, s$$
(8e)
(8f)

3) RENEWABLE ENERGY SOURCES

Equation (9a) limits the RES active power injection up to the forecasted available RES production. Equation (9b) relates the reactive power capability of RES with the injected active power. Equation (9c) gives the compensation the DSO pays to the RES owner, should curtailment be needed.

$$P_{w,t,s}^{Inj} \le P_{w,t,s}^{frc} \qquad \qquad \forall w, t, s \qquad (9a)$$

$$-0.33 \cdot P_{w,t,s}^{Inj} \le Q_{w,t,s}^{Inj} \le 0.33 \cdot P_{w,t,s}^{Inj} \quad \forall w, t, s$$
 (9b)

$$F_{w,t,s}^{RTD} = C_w^{curt} \left(P_{w,t,s}^{frc} - P_{w,t,s}^{lnj} \right) \qquad \forall w, t, s \qquad (9c)$$

C. INTERACTION OF DSO WITH THE TSO

In this framework, the DSO aims to stay committed to the hourly import/export schedule at the T-D, already informed to the TSO at noon a day in advance. Only if the reserved flexibility services from local DER have been exhausted, does the DSO import energy from the TSO. Equation (10a) gives the final import in RTD and (10b) calculates the cost imposed if deviations from the committed schedule occur at the T-D, i.e., the HV/MV SS.

$$P_{ss,t,s} = P_{ss,t}^{DAS} + P_{ss,t,s}^{Dev+} - P_{ss,t,s}^{Dev-} \quad \forall ss, t, s \quad (10a)$$

$$E^{RTD} = e^{Dev+} P^{Dev+} + e^{Dev-} P^{Dev-} \quad \forall ss, t, s \quad (10b)$$

 $F_{ss,t,s}^{KID} = \rho_{ss,t}^{Dev+} P_{ss,t,s}^{Dev+} + \rho_{ss,t}^{Dev-} P_{ss,t,s}^{Dev-} \forall ss, t, s \quad (10b)$ The penalties for upward and/or downward deviations are

set by the TSO and are related to the real-time system price and the cost of extra energy portions the TSO must dispatch to balance the system under its responsibility.

D. AC POWER FLOW MODEL

The AC power flow equations, given by (11a)–(11c), are applied $\forall i, t, s$ and $\forall j \in \Omega_{Ni}$:

$$P_{ij,t,s} = -V_{i,t,s}^2 G_{ij}$$

+ $V_{i,t,s} V_{j,t,s} \left(G_{ij} \cos \theta_{ij,t,s} + B_{ij} \sin \theta_{ij,t,s} \right)$ (11a)
 $O_{ij,t,s} = V_{i,t,s}^2 B_{ij}$

$$+ V_{i,t,s}V_{j,t,s} \left(G_{ij}\sin\theta_{ij,t,s} - B_{ij}\cos\theta_{ij,t,s}\right) \quad (11b)$$

$$\theta_{ij,t,s} = \theta_{i,t,s} - \theta_{j,t,s} \tag{11c}$$

The node power balance is expressed by (12a)–(12b):

$$\sum_{j\in\Omega_{Ni}} P_{ij,t,s} = \sum_{ss\in\Phi_i} P_{ss,t,s} + \sum_{w\in\Omega_{Wi}} P_{w,t,s}^{lnj} + \sum_{b\in\Omega_{Bi}} \left(P_{b,t,s}^{Dch} - P_{b,t,s}^{Ch} \right) + \sum_{ag\in\Omega_{AGi}} \left(P_{ag,t,s}^{A-FSD} - P_{ag,t,s}^{A-FSU} \right) - \sum_{d\in\Omega_{Di}} P_{d,t,s}^{frc} \quad \forall i, t, s \quad (12a)$$

$$\sum_{j\in\Omega_{Ni}} Q_{ij,t,s} = \sum_{ss\in\Phi_i} Q_{ss,t,s} + \sum_{w\in\Omega_{Wi}} Q_{w,t,s}^{Inj} + \sum_{ag\in\Omega_{AGi}} \left(Q_{ag,t,s}^{FSD} - Q_{ag,t,s}^{FSU} \right) - \sum_{d\in\Omega_{Di}} Q_{d,t,s}^{frc} \quad \forall i, t, s \quad (12b)$$

Equations (13a)–(13c) describe the system's technical constraints. Equation (13a) sets the bus voltage limits. Equations (13b) and (13c) set the capacity limits of network's power flows and HV/MV SS, respectively.

$$V^{\min} \le V_{i,t,s} \le V^{\max} \qquad \forall i, t, s \qquad (13a)$$

$$\left(P_{ij,t,s}\right)^{2} + \left(Q_{ij,t,s}\right)^{2} \le \left(S_{ij}^{\max}\right)^{2} \quad \forall i, t, s, j \in \Omega_{Ni}$$
(13b)

$$\left(P_{ss,t,s}\right)^2 + \left(Q_{ss,t,s}\right)^2 \le \left(S_{ss}^{\max}\right)^2 \quad \forall ss, t, s \tag{13c}$$

E. OBJECTIVE FUNCTION

The objective function of S-DAS, given by (14a), consists of two terms, each one related with the stage the decisions refer to. The first term is (14b) and the second term is the expected value of (14c), both over the 24-hour scheduling period, with 1 h resolution interval. Equation (14b) gives the cost of reserving flexibility services, and (14c) gives the cost of activating flexibility services under scenario s.

$$f^{S-DAS} = \underbrace{F^{DAS}\left(\mathbf{u}_{t}^{DAS}\right)}_{1^{st} \ stage} + \underbrace{\sum_{s \in S} \pi_{s} F_{s}^{RTD}\left(\mathbf{u}_{t,s}^{RTD}, \mathbf{x}_{t,s}^{RTD}\right)}_{2^{nd} \ stage}$$
(14a)

$$F^{DAS} = \sum_{t=1}^{I} \sum_{ag \in \Omega_{AG}} F^{DAS}_{ag,t} + \sum_{b \in \Omega_B} F^{DAS}_b$$
(14b)

$$F_{s}^{RTD} = \sum_{t=1}^{T} \left(F_{ss,t,s}^{RTD} + \sum_{ag \in \Omega_{AG}} F_{ag,t,s}^{RTD} + \sum_{w \in \Omega_{W}} F_{w,t,s}^{RTD} + \sum_{b \in \Omega_{B}} F_{b,t,s}^{RTD} \right)$$
(14c)

F. OVERALL FORMULATION

The general mathematical formulation of the S-DAS is given in (15) as follows:

$$\min f^{S-DAS}\left(\mathbf{u}_{t}^{DAS}, \mathbf{u}_{t,s}^{RTD}, \mathbf{x}_{t,s}^{RTD}\right)$$
(15a)

subject to

$$(5) - (14)$$
 (15b)

The proposed S-DAS problem is a NLP optimization problem and is solved using the state-of-the-art CONOPT4 commercial solver of GAMS [38]. In S-DAS, the duration of time-intervals is considered equal to one hour ($d_t = 1$), i.e., 24 time-intervals.

The S-DAS problem is inherently a large-scale optimization problem. Its computational complexity primarily depends on: a) the number of network buses, b) the multiple periods of the day-ahead planning and c) the number of the scenarios that model the uncertainties of wind speed and load. For instance, to optimize the day-ahead planning (24 hourly periods) for a 33-bus distribution system considering 9 states of wind speed and 9 states of load (81 scenarios of net load), the time elapsed is approximately 34 minutes. All tests in this paper were performed on a PC with an Intel Core i7-8700 CPU at 3.20 GHz and 8 GB of RAM. When the number of buses increases, the tractability degrades in a nonlinear fashion.

The computational performance could be improved by employing convex relaxations of the AC power flow equations, such as the semi-definite programming relaxation [39], the second order cone relaxation [40]–[41], and the quadratic convex relaxation [42].

In this paper, the authors' main interest is to propose a novel and effective S-DAS model to solve the problem and validate the extracted results, rather than improve the execution performance. Given the novelty of the proposed approach, it is paramount to ensure its reliability, as is done in this paper, before developing further add-ons.

IV. P-RTD FORMULATION

In this section, firstly a model to handle the response time of the FSP is introduced. Subsequently, the algorithm of the proposed P-RTD procedure is outlined based on the principles of MPC. Moreover, the overall P-RTD formulation is given.

A. HANDLING RESPONSE TIME

The time delay between receiving the activation signal sent by the DSO and achieving the requested output is defined as the response time of the flexibility service. The response time is a critical parameter during RTD as it plays significant role in the optimal flexibility sources management and the efficient activation of the less expensive flexibility services on time. Response time varies from a few seconds to few minutes and depends on the technologies used to implement the flexibility services, as analyzed in [24] and [25].



FIGURE 2. Handling response time when equal to one timeslot of the P-RTD procedure.

Let define $\mathbf{u}_{t+h|t}$ as the vector of control actions with response time equal to one timeslot during the h – th timeslot of the t – th rolling horizon, where h = 0...H - 1. The handling of the response time during P-RTD can be depicted by Fig. 2. Whenever the response time is equal to the timeslot of the dispatch procedure, the vector of the current control actions $\mathbf{u}_{t_n|t_n}$ cannot be immediately dispatched and is replaced by the vector of the control actions $\mathbf{u}_{t_{n+1}|t_{n-1}}$ of the second timeslot of the previous rolling horizon. In case the response time is equal to twice the timeslot of the dispatch procedure, the vector of the current control actions $\mathbf{u}_{t_n|t_n}$ is replaced by the control actions $\mathbf{u}_{t_{n+2}|t_{n-1}}$ of the third timeslot of the previous rolling horizon, and so on. For technologies with response time quite faster than the timeslot of the dispatch procedure, it is assumed that the vector of the current control actions $\mathbf{u}_{t_n|t_n}$ is applied at the beginning of the current timeslot.

BSS are characterized by a fast response time, up to some seconds, and can quickly compensate mismatches attributed

to short-term forecasting errors. RES curtailment can be activated with response time up to some seconds, but the response time may be set to longer durations (e.g., some minutes) depending on system operator rules to avoid unwilling ramp-up production, e.g., due to wind gusts. Here, RES maximum permitted production is announced 15 minutes in advance. As far as the response time of DERA is concerned, the activation signal is usually sent from 5 to 30 minutes in advance depending on the energy technologies used and their contractual agreement with the DSO. In this paper, a response time of 15 minutes is assumed, which is a prerequisite for many DSOs when procuring flexibility services [43].

B. PROPOSED P-RTD BASED ON MPC PRINCIPLES

In general, MPC allows for the optimization of current timeslots, while always taking future time-slots into account, by constantly re-predicting future events based on the actual flow of real events. This is achieved by continuously optimizing the system's operation under a finite horizon but implementing the control actions only on the current operating condition, thus performing a brand-new optimization for every single operating condition, under an ever-changing rolling time horizon.

In the proposed P-RTD, the DSO solves a look-ahead multi-period OPF problem for each time-interval. The optimization depends on the current value of the system's dynamic state variables, i.e., BSS state of charge at the end of the previous time-interval and the DERA's set-point of the previous time-interval due to ramping limits. The procedure of the proposed P-RTD is implemented as follows:

a) at time-interval *t* the multi-period OPF problem is solved over a predefined rolling horizon with *H* time-intervals. *T* he control variables $\mathbf{u}_{t+h|t}$ are computed for h = 0...H-1, where $\mathbf{u}_{t|t}$ is the vector of current values of the control variables of the time-interval *t* and $\mathbf{u}_{t+h|t}$ is the vector of the values of the control variables for the time-interval t+h, given the system's state at time-interval *t*.

b) the P-RTD implements: i) the currently optimized values for h = 0, i.e., $\mathbf{u}_{t|t}$, these ones with faster response time (up to 1 min), and ii) the values optimized for the second time-interval during the previous rolling horizon for h = 1, i.e., $\mathbf{u}_{t+1|t-1}$, these ones with response time equal to the timeinterval t.

c) the actual values (received from measurement units), the updated short-term forecasting values and the values of the system's state are obtained and are fed back to the procedure.

d) the optimization process is repeated from step a) for the next time-interval t = t+1 having new short-term forecasting values for the next rolling horizon.

Based on the above procedure, the control actions of BSS charging/discharging and DERA's FSU and FSD are constantly updated to minimize the objective function of (18a) over the finite rolling horizon by considering: a) the most recent forecasted data, b) the control actions of the flexibility

sources with greater response time, and c) the current state of the system.

C. P-RTD OVERALL FORMULATION

Equations (6), (8)–(13) are used in this stage by excluding the subscript *s* (where it exists) and by replacing the subscript *t* with t + h|t that indicates the variable of the h – th time-interval of the t – th rolling horizon. Equation (9a) is reformulated with (16a)–(16b), where $P_{w,t+1|t-1}^{Inj}$ is the set-point (maximum permitted injection) of RES *w* for h = 1 that was optimized in the previous rolling horizon (t - 1).

$$P_{w,t+h|t}^{Inj} \le P_{w,t+h|t}^{frc} \qquad \forall w, t, h \neq 0 \qquad (16a)$$

$$P_{w,t+h|t}^{Inj} = \min(P_{w,t+1|t-1}^{Inj}, P_{w,t}^{act}) \quad \forall w, t, h = 0$$
(16b)

In P-RTD, (14c) is modified to formulate the objective function of the t – th rolling horizon, as follows:

$$f_{t+h|t}^{P-RTD} = \sum_{h=0}^{H-1} \left(F_{ss,t+h|t}^{RTD} + \sum_{ag \in \Omega_{AG}} F_{ag,t+h|t}^{RTD} + \sum_{w \in \Omega_W} F_{w,t+h|t}^{RTD} + \sum_{b \in \Omega_B} F_{b,t+h|t}^{RTD} \right)$$
(17)

The general mathematical formulation of the multi-period OPF problem of the t – th rolling horizon is given by (18).

$$\min f_{t+h|t}^{P-RTD} \left(\mathbf{u}_{t+h|t}^{P-RTD}, \mathbf{x}_{t+h|t}^{P-RTD} \right)$$
(18a)

subject to

$$(6), (8), (9a) - (9c), (10) - (13), (16) - (17)$$
 (18b)

The iterative procedure of P-RTD has been coded in MATLAB, while the optimization model (formulated as a NLP problem) is solved using the state-of-the-art CONOPT4 commercial solver of GAMS [38]. In P-RTD, the duration of time-intervals is considered less than one hour with $d_t \in (0, 1]$, e.g., if the duration is 15 min, $d_t = 0.25$ and the number of rolling horizons over a day is 96.

In contrast with the computational complexity of S-DAS, the P-RTD of each time-interval is executed in less than 10 seconds. This is mainly due to the absence of the scenarios of uncertainty in P-RTD.

V. CASE STUDIES & NUMERICAL RESULTS

This section is dedicated to investigating the DSO's level of risk exposure by applying the proposed flexibility services management framework. At the very beginning of this section, the methodology that evaluates the risk exposure is introduced. Subsequently, the proposed flexibility services management framework is implemented in two case studies. The first is based on a 4-bus distribution system and is presented as a proof-of-concept validation. The second is based on a 33-bus distribution system and is presented for further investigation.

TABLE 2. Cases of risk exposure and their corresponding rate $1 - \beta$.

	Prediction inte		Rate of risk			
Case	Coverage rates α_L and α_W (%)	Upper and lower bounds	States of Table 1 considered	Number of scenarios	exposure 1–β (%)	
А	99.7	$\hat{y}_{t+h} \pm 3\sigma_h$	1, 2, 3, 4, 5, 6, 7, 8, 9	81	0.6	
В	95.5	$\hat{y}_{t+h} \pm 2\sigma_h$	2, 3, 4, 5, 6, 7, 8	49	8.8	
С	86.7	$\hat{y}_{t+h} \pm 1.5\sigma_h$	3, 4, 5, 6, 7	25	24.8	
D	68.3	$\hat{y}_{t+h} \pm 1\sigma_h$	4, 5, 6	9	53.3	

A. RISK EXPOSURE EVALUATION METHODOLOGY

1) CASES OF RISK EXPOSURE

Four different cases (presented in Table 2) are considered in order to investigate the DSO's level of risk exposure. Each case of risk exposure is linked to a certain prediction interval of load (α_L) and a certain prediction interval of wind speed (α_V), the coverage rates of which are considered equal, as presented in the second column of Table 2. The third column of Table 2 gives the prediction interval bounds, for both load and wind speed, following the notation of (1) and assuming that both errors are normally distributed with zero mean. The prediction intervals of load and wind speed are selected based on the states of Table 1, which are shown in the fourth column of Table 2. Pertaining to the last column, the rate of risk exposure is $1 - \beta$, with $\beta \in [0, 1]$, where β is the coverage rate of the selected scenarios as defined in (4d) of Section III.A.4. The selection of the cases of Table 2 is based on the modelling of Section III.A.

Case A is the most conservative among the selected cases with a rate of risk exposure $(1-\beta)$ of only 0.6%, which means virtually zero risk exposure for the DSO. To create the scenarios of Case A, all the states of load forecast error and wind speed forecast error that are shown in Table 1 of Section III.A are included, the combination of which concludes in 81 scenarios, each with its own probability of occurrence.

On the other hand, Case B, Case C and Case D demonstrate the DSO's level of risk exposure given that the higher the rate of risk exposure $(1 - \beta)$ is, the less the reserves are, and thus the riskier the RTD is. Therefore, Case D is the least conservative among the selected cases, because it has a rate of risk exposure $(1 - \beta)$ of 53.3%. Transitioning from Case A to Case B, the states that lie on the tails of the forecast error's normal distribution, i.e., those with smaller probability (1st and 9th states) are excluded. Combining the 7 remaining states of load forecast error with those 7 remaining states of wind speed forecast error, the 49 scenarios of Case B are created. Their probabilities are normalized to neglect the probabilities of the rejected scenarios in S-DAS, following the theory of the truncated normal distribution applied in [44]. Following the same procedure and excluding more states, Case C and Case D comprise 25 and 9 scenarios, respectively.

2) RISK EXPOSURE METRIC

The metric *additional risk exposure*, denoted by $ARE_{X \rightarrow Y}$, is introduced to measure the risk exposure, when transitioning from Case X to Case Y, given that Case X is more conservative (has lower rate of risk exposure) than Case Y. As shown in (19), *ARE* is the ratio of the increase in RTD cost over the reduction of DAS cost, when transitioning from Case X to Case Y. In fact, *ARE* is a ratio that measures the increase in RTD cost as a result of reducing the DAS cost by an additional monetary unit (e.g., \in).

$$ARE_{X \to Y} = \frac{RTD_{X \to Y}}{DAS_{X \to Y}} = \frac{RTD_Y - RTD_X}{DAS_X - DAS_Y}$$
(19)

where RTD_X and DAS_X denote the RTD cost and the DAS cost of Case X, respectively.

The values of *ARE* can be interpreted as follows. If $ARE_{X \to Y}$ has a value less than unit, the transition from Case X to Case Y is beneficial for the DSO, since the DAS cost reduction is higher than the RTD cost increase. Inversely, if the value of $ARE_{X \to Y}$ is greater than unit, it is no worth transitioning from Case X to Case Y.



FIGURE 3. 4-bus distribution system (illustrative example).

B. CASE STUDY OF 4-BUS DISTRIBUTION SYSTEM: AN ILLUSTRATIVE EXAMPLE

1) CHARACTERISTICS OF THE DISTRIBUTION SYSTEM

The 4-bus distribution system of Fig. 3 is used in a proofof-concept framework. This network is radial, with nominal voltage of 20 kV, and is supplied by an 8 MVA capacity HV/MV SS. The maximum line capacity is set at 8 MVA and the peak load is equal to 3.68 MVA. The net load of bus 2 is regulated by a DERA, offering maximum FSU of 0.4 MW with energy capacity of 8 MWh per day and maximum FSD of 0.8 MW with energy capacity of 15 MWh per day. The ramping limits (ramp-up and ramp-down) are set to 0.4 MW/h and 0.8 MW/h for FSU and FSD, respectively. Multiple DER are also in place: a WG of 4 MW installed capacity, a PV station of 0.6 MWp, and a grid-scale BSS with 0.5 MW, 2.0 MWh, and 90% round-trip efficiency.

Figs. 4 and 5 depict the probabilistic forecasts of network's total load and wind generation, respectively, with their corresponding prediction intervals (shadowed areas) in accordance with the cases of Table 2. The point forecasts and the actual measurements are also shown to illustrate the actual deviations from the day-ahead point forecasts.

With regard to the remuneration schemes described in Section II.B, Table 3 shows the prices concerning FSP



FIGURE 4. Network load: actual values, day-ahead point forecasts and the corresponding prediction intervals (4-bus distribution system).



FIGURE 5. Wind generation: actual values, day-ahead point forecasts and the corresponding prediction intervals (4-bus distribution system).

TABLE 3. Remuneration for the FSP of the 4-bus distribution system.

FSP	Commitment price	Activation price
DERA FSD & FSU	€20/MW-h	€40/MWh
BSS	€180/MW-day	€10/MWh
RES curtailment	-	€120/MWh

participation in this test system. The penalties for deviating from the import schedule at the T-D are given in Fig. 6 (the penalties of upward deviation are considered equal to those of downward deviation).

For the day-ahead forecasting, the standard deviation for the load forecast error and that for the wind speed forecast error are considered equal to 10% and 15%, respectively, of the point forecasts [16]. It is usual the wind speed forecast error to be greater than that of the load forecast [45].

As far as the short-term forecasting is concerned, the predictor of the P-RTD procedure is assumed to have an error with standard deviation that varies over time, from 3%(1 h ahead) to 15% (24 h ahead) for the wind speed, and from 1% (1 h ahead) to 10% (24 h ahead) for the load forecast (expressed as percentage of the point forecasts).



FIGURE 6. T-D deviation penalties imposed by the TSO based on the day-ahead market price.

TABLE 4. Summary of S-DAS results for the 4-bus distribution syste	em.
--	-----

	DAS stage			RTD stage (expected cost values)				
	Reserves			T-D	Activation		WG	
Case	DERA (E)	BSS (E)	Total (€)	devia- tions (E)	DERA (E)	BSS (€)	curtail- ment (E)	Total (€)
А	455.9	90.0	545.9	614.5	355.5	23.4	187.2	1,180.6
В	415.1	90.0	505.1	425.5	313.8	20.2	135.6	895.1
С	309.4	90.0	399.4	381.4	222.1	16.2	122.4	742.1
D	191.1	53.9	245.0	251.3	132.5	10.4	96.9	491.1

2) S-DAS RESULTS

The results of S-DAS, presented in Table 4, show that the more conservative the DSO is, the higher the reserve costs become. The total DAS cost of Case A (€545.9) with rate of risk exposure of only 0.6% is more than twice as expensive compared to the total DAS cost of Case D (≤ 245.0) with rate of risk exposure of 53.3%. Most of the cost that is avoided by adopting riskier approaches is attributed to the significant reduction in the level of DERA reserves, while the reservation costs of the remaining flexibility options, i.e., the BSS, remain relatively at the same level, as Table 4 shows. This makes DERA the additional option in guaranteeing system balance, as well as the source for most of the DSO's dayahead expenses. Fig. 7 shows how the total energy of FSD is allocated across the day, making clear that when considering Case C and Case D for S-DAS, the reserved FSD energy is much lower than that of considering Case A and Case B. The narrower the coverage rates of the prediction intervals of load and wind speed are, the less the expected deviations from the point forecasts are supposed to be, and thus lower FSD reserves are needed.

3) P-RTD RESULTS

Using the reserves scheduled with the S-DAS, the DSO applies the proposed P-RTD to stay committed to the hourly import schedule at the T-D, while exploiting the available flexibility services with the least possible expenses. The P-RTD is executed with time-resolution of 15 minutes and is tested for different rolling horizons to highlight the importance of look-ahead optimization in RTD.



FIGURE 7. DERA's reserved FSD considering different cases of risk exposure for S-DAS (4-bus distribution system).



FIGURE 8. RTD cost for different rolling horizons considering different cases of risk exposure for S-DAS (4-bus distribution system).

Fig. 8 illustrates the RTD cost for different look-ahead rolling horizons considering different cases of risk exposure when solving the S-DAS. In the bar-chart of Fig. 8, the first set of bars, labeled with "no P-RTD", refers to an ordinary RTD strategy that does not consider look-ahead horizon. The set of bars "no P-RTD" is used as benchmark for comparison. The rest sets of bars refer to the length of the selected rolling horizon of the P-RTD, e.g., the label "4h P-RTD" denotes P-RTD of a 4 h rolling horizon. Fig. 8 shows that in all cases of risk exposure, the proposed P-RTD has better performance (lower RTD cost) than "no P-RTD". The longer the rolling horizon of the P-RTD is, the lower the RTD cost is.

Depending on the length of the rolling horizon, the P-RTD process differs in accuracy (the extension of the rolling horizon increases the forecast errors) and thus different costs are incurred. For instance, it is observed from Fig. 8 that the RTD cost of Case A is reduced from €649.5 ("no P-RTD") to €555.6 ("1h P-RTD"), i.e., a reduction of 14.5% from "no P-RTD"; when extending the rolling horizon by 1 hour the RTD cost of Case A is reduced from €555.6 ("1h P-RTD") to €527.1 ("2h P-RTD"), i.e., an additional reduction of 5.1%; and when extending the rolling horizon by 2 hours ("4h P-RTD") the additional reduction from "2h P-RTD" to "4h P-RTD" is about 1%. While the extension of the lookahead horizon generally reduces the costs incurred, the additional reduction quickly diminishes. Extending the rolling horizon's length makes sense until a certain point. After a certain threshold is reached, it appears that the marginal gains are minimal. In fact, if the P-RTD approach is also coupled with

a very broad look-ahead horizon, it may lead to additional costs, due to the fact that the more extended is the horizon the higher is the variance of the forecast error. Thus, apart to how much value the DSO gains from considering prediction intervals with higher coverage rate for S-DAS, there is also a case to be made on how much value the DSO gains from extending the rolling horizon of P-RTD.

There is merit to take a closer look at Fig. 8 and analyze the reduction of RTD cost for different cases of risk exposure, while keeping the length of the rolling horizon to a fixed value. For example, focusing on the set of bars "4h P-RTD", the RTD cost of Case A is reduced from €649.5 ("no P-RTD") to €520.1, i.e., a reduction of 19.9% from that of "no P-RTD"; the RTD cost of Case B (equal to that of Case A) is also reduced by 19.9%, while the RTD cost of Case C and Case D are reduced by 17.5% and 10.9%, respectively, from those of "no P-RTD". This shows that the performance of P-RTD, not only depends on the rolling horizon selected, but also depends on the flexibility services reserved, highlighting the dynamic nature of the P-RTD.



FIGURE 9. Import profile at the T-D considering Case C for S-DAS.



FIGURE 10. DERA's activated FSD considering Case C for S-DAS.

Figs. 9, 10 and 11 refer to P-RTD of 4 h rolling horizon ("4h P-RTD") and present results of RTD when considering Case C for the S-DAS. The results of Figs. 9, 10 and 11 are compared with the results obtained if a dispatch strategy without predictive control ("no P-RTD") is implemented. Fig. 9 illustrates the import at the T-D and compares the curves of "no P-RTD" and "4h P-RTD" with that scheduled during DAS. This comparison determines the deviations of actual active power flows from the committed ones at the T-D. Fig. 10 depicts the FSD activation that the DERA



FIGURE 11. BSS activation considering Case C for S-DAS.

provides, while Fig. 11 presents the operation of the BSS by depicting its state of charge. By simultaneously analyzing Figs. 9, 10 and 11, one can notice that P-RTD ends up in lower deviations during the time window 16:00–21:00 (between 64-th and 84-th time-interval), when T-D deviation penalties (Fig. 6) are much higher, than during early morning hours.

4) DAS & RTD OVERALL DISCUSSION

The results extracted for the illustrative example (Sections V.B.2 and V.B.3) signify the vital need for careful S-DAS. In the pursuit of reducing the reserve costs, the adopted risk may be very high, and may lead to hefty intraday balancing costs that are not offset by the savings of the DAS stage, regardless of how sophisticated the utilized real-time prediction tool is.

Fig. 12 presents the total cost breakdown in RTD cost and DAS cost for the four cases of risk exposure defined in Table 2 of Section V.A. In Fig. 12, the RTD cost refers to P-RTD simulation with a 4 h rolling horizon. Table 5 gives the values of *ARE* metric (defined in Section V.A.2). As an example, the *ARE* for the transition from Case B to Case C (denoted as $B \rightarrow C$ in Table 5) is computed as follows:

$$ARE_{B\to C} = \frac{RTD_C - RTD_B}{DAS_B - DAS_C} = \frac{551.40 - 520.10}{505.10 - 399.40}$$
$$= \frac{31.30}{105.70} = 0.30$$

The $ARE_{A\to B} = 0.00$ (Table 5) shows that the transition from Case A to Case B is beneficial for the DSO, because no extra RTD cost is incurred as a result of the additional gains achieved through DAS cost reduction by $\in 40.8$. Increasing once more the rate of risk exposure $(1 - \beta)$ and transitioning from Case B to Case C, $ARE_{B\to C} = 0.30$. This indicates that the B \rightarrow C transition reduces the DAS cost more than three times compared to the RTD cost increase, as is also shown in Table 5. Consequently, the transition B \rightarrow C is also beneficial for the DSO.

The $ARE_{C \rightarrow D} = 2.03$ (Table 5) shows that the transition from Case C to Case D is not beneficial for the DSO. Indeed, although the DAS cost is very low in Case D (Fig. 12), the DSO has severely jeopardized system balance, and as a result, the RTD cost of Case D is much higher than that of Case C (Fig. 12). The risk undertaken in Case D (much higher RTD cost) is not properly compensated by the DAS cost reduction. Keeping less reserves in Case D causes much

	$DAS_{X \to Y}$	$RTD_{X \to Y}$	
Case transition	cost reduction	cost increase	ARE
$A \rightarrow B$	40.8	0.0	0.00
$B \rightarrow C$	105.7	31.3	0.30
$C \rightarrow D$	154.4	312.8	2.03
520.10 €			
520.10€	520.10€	551.40€	864.20€

TABLE 5. Risk exposure metric (ARE) for the 4-bus distribution system for

Case A Case B Case C Case D

FIGURE 12. Total cost (DAS and RTD cost) considering different cases of risk exposure for S-DAS (4-bus distribution system).

TABLE 6. Remuneration for the FSP of the 33-bus distribution system.

FSP	Commitment price	Activation price
DERA-1 FSD & FSU	€20/MW-h	€40/MWh
DERA-2 FSD & FSU	€25/MW-h	€30/MWh
BSS	€180/MW-day	€10/MWh
RES curtailment	-	€120/MWh

higher T-D deviations (from the schedule), which are accompanied by stiff penalties (those of Fig. 6), as well as more expensive flexibility activation.

C. CASE STUDY OF 33-BUS DISTRIBUTION SYSTEM 1) CHARACTERISTICS OF THE DISTRIBUTION SYSTEM

The 33-bus distribution system of [46], modified according to Fig. 13, is also studied. The basic network data are directly sourced from [46]. This network is radial, with nominal voltage of 12.66 kV, and is supplied by a 10 MVA capacity HV/MV substation. The maximum line capacity is set at 8.7 MVA and the peak load is considered equal to 4.37 MVA. The network contains controllable loads by DERA, and also has multiple installed DER: WG, PV stations, and grid-scale BSS. The WG total installed capacity is 3.2 MW and the PV station's installed capacity is 0.6 MWp. Two grid-scale BSS of 1.3 MW total power capability and 2.5 MWh capacity (90% round-trip efficiency) are also part of the system.

Two DERA are competing in this case study, by regulating the net load of different buses: DERA-1 controls buses 8, 14 and 31, while DERA-2 controls bus 25. DERA-1 offers FSU of 0.6 MW maximum power and energy capacity of 11 MWh/day, as well as FSD of equal power and energy capacity. DERA-2 offers FSU of 0.3 MW maximum power with energy capacity of 6 MWh and FSD of equal power and capacity. The ramping limits (ramp-up and ramp-down) of DERA-1 and DERA-2 are set to 1.2 MW/h and 0.6 MW/h,



FIGURE 13. Modified 33-bus distribution system.



FIGURE 14. Total cost (DAS and RTD cost) considering different cases of risk exposure for S-DAS (33-bus distribution system).

 TABLE 7. Risk exposure metric (ARE) for the 33-bus distribution system

 for P-RTD of 4h rolling horizon.

Case transition	$DAS_{X \to Y}$ cost reduction (€)	$RTD_{X \to Y}$ <i>cost increase</i> (€)	ARE
$A \rightarrow B$	67.10€	38.30€	0.57
$B \rightarrow C$	118.20€	70.10 €	0.59
$C \rightarrow D$	109.40 €	180.20 €	1.65

accordingly. The prices remunerated to the FSP that participate in this case study are given in Table 6.

2) RESULTS

Fig. 14 shows the breakdown of total cost for the four cases of risk exposure presented in Table 2 of Section V.A, for the 33-bus distribution system. Fig. 14 shows that Case C results in an overall cost (DAS and RTD cost) of \in 971.72, which is the lowest value among the selected cases. Even though the rate of risk exposure is higher when deciding to solve S-DAS with Case C, the DSO reduces its overall cost by 7.3% in comparison with that of Case A (\in 1,048.62), which is the most conservative case. Once again, there is a threshold of risk exposure, over which the DSO's total expenses begin to increase. Indeed, as is shown in Table 7, the values of *ARE* metric are less than unit for the transitions A \rightarrow B and B \rightarrow C, while *ARE* is greater than unit for the transition C \rightarrow D.

Interesting findings can be drawn when multiple DERA are participating in the local flexibility market. Fig. 15 shows the reserved FSU and FSD versus those activated by the



FIGURE 15. Reserved versus activated FSU and FSD per DERA considering Case C for S-DAS (33-bus distribution system).

two DERA participating in the 33-bus distribution system, i.e., DERA-1 and DERA-2. The results of Fig. 15 refer to the reserves of FSU and FSD considering Case C for the S-DAS. The activation of FSU and FSD considers a 4 h rolling horizon for P-RTD. As Fig. 15 shows, the DSO activates more the FSD of DERA-1 (4.13 MWh) than that of DERA-2 (0.45 MWh). However, with a closer inspection of Fig. 15, the degree the DSO exploits the reserved FSD of DERA-2 (63% of the reserved FSD, which is 0.72 MWh) is higher than that of DERA-1 (48% of the reserved FSD). This is reasonable, because the activation price the DERA-2 has bidden is cheaper than that of DERA-1. Of course, DERA-1 FSD total energy is higher, as the energy bid is also much higher than that of DERA-2. However, when it comes to FSU, even though DERA-2 activation price is still lower, the FSU reserves are zero owing to the higher commitment price $(\in 25/MW-h)$ compared to that of DERA-1 $(\in 20/MW-h)$, and thus no FSU could be activated by DERA-2.

D. OVERALL DISCUSSION

There is a clear trade-off between the reserved flexibility services and the real-time expenses. In other words, sacrificing a part of the reserves of Case A, the DSO spends less in total (DAS plus RTD expenses). What is more important is to investigate the maximum level of risk exposure, i.e., the portion of the reserves that could be sacrificed leading to lower total expenses. This varies among different test systems and depends on the cost of reserves, the cost of activation and the penalties of deviating from the scheduled power flows at the T-D. Further investigation is needed in this direction, in order to be able to elaborate more the results of different test systems for different range of net load scenarios to draw more specific conclusions for the flexibility services management under uncertainties.

VI. CONCLUSION AND FUTURE WORK

This paper introduces a flexibility services management framework consisting of two optimization stages: 1) a stochastic scheduling approach to efficiently handle both dayahead uncertainties and intra-day fluctuations, and 2) a realtime dispatch based on predictive control used as validation of the DAS. The work addresses contemporary issues for both optimization stages that are frequently disregarded. The general insufficiency of the deterministic day-ahead scheduling is demonstrated due to the always present forecasting errors. Even so, the inclusion of stochastic programming to model the planning stage can prove inadequate, leading to over- or underestimations of the planning decision due to the high variability of renewable technologies. In addition, the dire need for a dynamic and highly adaptable approach for real-time dispatch applications is demonstrated by employing the MPC principles. The constantly updated short-term predictions on the output of wind farms are such that require constant monitoring with a continuous stream of updated setpoints timely sent to the various rapidly responding technologies (e.g., BSS), which are a powerful tool in the DSO's arsenal.

With respect to the recorded numerical results, it has been shown that when performing stochastic programming, there is a limit to the level of risk exposure, i.e., what range of scenarios should be considered, when compared to the marginal gains, especially if they are not high enough to offset the extra RTD cost. While additional reserved flexibility is certainly not undesirable, approaches that are overly conservative are generally avoided in the industry sector, as they usually lead to more expensive solutions. The limits of risk exposure with respect to marginal gains are illustrated by the various values of *ARE* metric that is introduced to measure the DSO's level of risk exposure. It is also shown that when performing optimization for real-time dispatch based on a look-ahead technique, there is a saturation to the effectiveness of extending the time horizon.

The handling the response time of the FSP is also a critical aspect, which is illustrated in this paper. It is the first time that a model is formulated to effectively activate the less expensive flexibility services on time, while considering a look-ahead technique. The quantification of the benefits from handling the response time is also valuable, but not within the scope of this paper; however it is strongly recommended as a future work.

Moreover, an issue not within the scope of the paper but with significant research interest is the tractability of the proposed approach. While proven effective, its reliable application in much larger distribution systems is challenging. As such, the proper approximations must be devised to expand the applicability of the proposed approach for larger problem sizes. Furthermore, an issue that has been ignored by past research works is the case of the TSO making emergency requests for flexibility from the DSO. The approach that the DSO would have to adopt is to include these new unplanned (time-wise and quantity-wise) parameters in its schedule, and it is an issue that has not been properly addressed in the scientific literature. It is in that light that the authors plan to significantly extend their work by not only incorporating even more modern aspects of active distribution networks, but also doing so in a scalable fashion.

REFERENCES

- H. Farhangi, "The path of the smart grid," *IEEE Power Energy Mag.*, vol. 8, no. 1, pp. 18–28, Jan./Feb. 2010.
- [2] L. Kane and G. W. Ault, "Evaluation of wind power curtailment in active network management schemes," *IEEE Trans. Power Syst.*, vol. 30, no. 2, pp. 672–679, Mar. 2015.
- [3] P. Siano and D. Sarno, "Assessing the benefits of residential demand response in a real time distribution energy market," *Appl. Energy*, vol. 161, pp. 533–551, Jan. 2016.
- [4] W. Liu, Q. Wu, F. Wen, and J. Ostergaard, "Day-ahead congestion management in distribution systems through household demand response and distribution congestion prices," *IEEE Trans. Smart Grid*, vol. 5, no. 6, pp. 2739–2747, Nov. 2014.
- [5] I. I. Avramidis, V. A. Evangelopoulos, P. S. Georgilakis, and N. D. Hatziargyriou, "Demand side flexibility schemes for facilitating the high penetration of residential distributed energy resources," *IET Gener.*, *Transmiss. Distrib.*, vol. 12, no. 18, pp. 4079–4088, Oct. 2018.
- [6] F. Capitanescu and I. Bilibin, "A tractable two-step MILP–QCP approach to on-line thermal constraint management in large radial active distribution systems," *Electric Power Syst. Res.*, vol. 140, pp. 580–587, Nov. 2016.
- [7] A. Vicente-Pastor, J. Nieto-Martin, D. W. Bunn, and A. Laur, "Evaluation of flexibility markets for Retailer-DSO-TSO coordination," *IEEE Trans. Power Syst.*, vol. 34, no. 3, pp. 2003–2012, May 2019.
- [8] V. A. Evangelopoulos, P. S. Georgilakis, and N. D. Hatziargyriou, "Optimal operation of smart distribution networks: A review of models, methods and future research," *Electr. Power Syst. Res.*, vol. 140, pp. 95–106, Nov. 2016.
- [9] CEER, Brussels, Belgium. (2016). Position Paper on the Future DSO and TSO Relationship. [Online]. Available: https://www.ceer.eu/1305
- [10] Towards Smarter Grids: Developing TSO and DSO Roles and Interactions for the Benefit of Consumers, ENTSO-E, Brussels, Belgium, 2015.
- [11] L. N. Ochoa, F. Pilo, A. Keane, P. Cuffe, and G. Pisano, "Embracing an adaptable, flexible posture: Ensuring that future European distribution networks are ready for more active roles," *IEEE Power Energy Mag.*, vol. 14, no. 5, pp. 16–28, Sep. 2016.
- [12] H. Gerard, E. I. Rivero Puente, and D. Six, "Coordination between transmission and distribution system operators in the electricity sector: A conceptual framework," *Utilities Policy*, vol. 50, pp. 40–48, Feb. 2018.
- [13] A. Gabash and P. Li, "Active-reactive optimal power flow in distribution networks with embedded generation and battery storage," *IEEE Trans. Power Syst.*, vol. 27, no. 4, pp. 2026–2035, Nov. 2012.
- [14] S. Gill, I. Kockar, and G. W. Ault, "Dynamic optimal power flow for active distribution networks," *IEEE Trans. Power Syst.*, vol. 29, no. 1, pp. 121–131, Jan. 2014.
- [15] S. Golshannavaz, S. Afsharnia, and F. Aminifar, "Smart distribution grid: Optimal day-ahead scheduling with reconfigurable topology," *IEEE Trans. Smart Grid*, vol. 5, no. 5, pp. 2402–2411, Sep. 2014.
- [16] A. Saint-Pierre and P. Mancarella, "Active distribution system management: A dual-horizon scheduling framework for DSO/TSO interface under uncertainty," *IEEE Trans. Smart Grid*, vol. 8, no. 5, pp. 2186–2197, Sep. 2017.
- [17] W. A. Bukhsh, C. Zhang, and P. Pinson, "An integrated multiperiod OPF model with demand response and renewable generation uncertainty," *IEEE Trans. Smart Grid*, vol. 7, no. 3, pp. 1495–1503, May 2016.

- [18] T. Soares, R. J. Bessa, P. Pinson, and H. Morais, "Active distribution grid management based on robust AC optimal power flow," *IEEE Trans. Smart Grid*, vol. 9, no. 6, pp. 6229–6241, Nov. 2018.
- [19] L. Xie and M. D. Ilic, "Model predictive dispatch in electric energy systems with intermittent resources," in *Proc. IEEE Int. Conf. Syst., Man Cybern.*, Singapore, Oct. 2008, pp. 42–47.
- [20] E. Mayhorn, K. Kalsi, M. Elizondo, W. Zhang, S. Lu, N. Samaan, and K. Butler-Purry, "Optimal control of distributed energy resources using model predictive control," in *Proc. IEEE Power Energy Soc. Gen. Meeting*, Jul. 2012, pp. 1–8.
- [21] P. P. Zeng, Z. Wu, X.-P. Zhang, C. Liang, and Y. Zhang, "Model predictive control for energy storage systems in a network with high penetration of renewable energy and limited export capacity," in *Proc. Power Syst. Comput. Conf.*, Wroclaw, Poland, Aug. 2014, pp. 1–7.
- [22] M. Arnold and G. Andersson, "Model predictive control for energy storage including uncertain forecasts," in *Proc. 17th Power Syst. Comput. Conf.* (*PSCC*), Stockholm, Sweden, Aug. 2011, pp. 24–29.
- [23] W. Su, J. Wang, K. Zhang, and A. Q. Huang, "Model predictive controlbased power dispatch for distribution system considering plug-in electric vehicle uncertainty," *Electr. Power Syst. Res.*, vol. 106, pp. 29–35, Jan. 2014.
- [24] B. Mohandes, M. S. E. Moursi, N. Hatziargyriou, and S. E. Khatib, "A review of power system flexibility with high penetration of renewables," *IEEE Trans. Power Syst.*, vol. 34, no. 4, pp. 3140–3155, Jul. 2019.
- [25] O. Ma, N. Alkadi, P. Cappers, P. Denholm, J. Dudley, S. Goli, M. Hummon, S. Kiliccote, J. MacDonald, N. Matson, D. Olsen, C. Rose, M. D. Sohn, M. Starke, B. Kirby, and M. O'Malley, "Demand response for ancillary services," *IEEE Trans. Smart Grid*, vol. 4, no. 4, pp. 1988–1995, Dec. 2013.
- [26] C. Eid, P. Codani, Y. Perez, J. Reneses, and R. Hakvoort, "Managing electric flexibility from distributed energy resources: A review of incentives for market design," *Renew. Sustain. Energy Rev.*, vol. 64, pp. 237–247, Oct. 2016.
- [27] H. Klinge Jacobsen and S. T. Schröder, "Curtailment of renewable generation: Economic optimality and incentives," *Energy Policy*, vol. 49, pp. 663–675, Oct. 2012.
- [28] M. N. I. Sarkar, L. G. Meegahapola, and M. Datta, "Reactive power management in renewable rich power grids: A review of grid-codes, renewable generators, support devices, control strategies and optimization algorithms," *IEEE Access*, vol. 6, pp. 41458–41489, 2018.
- [29] P. Pinson and G. Kariniotakis, "Conditional prediction intervals of wind power generation," *IEEE Trans. Power Syst.*, vol. 25, no. 4, pp. 1845–1856, Nov. 2010.
- [30] J. M. Morales, A. J. Conejo, H. Madsen, P. Pinson, and M. Zugno, *Integrating Renewables in Electricity Markets-Operational Problems*, New York, NY, USA: Springer, 2014, pp. 21–26.
- [31] S. G. Makridakis, S. C. Wheelwright, and R. J. Hyndman, *Forecasting: Methods and Applications*, 3rd ed. New York, NY USA: Wiley, 1997, pp. 52–54.
- [32] A. Khosravi, S. Nahavandi, and D. Creighton, "Prediction intervals for short-term wind farm power generation forecasts," *IEEE Trans. Sustain. Energy*, vol. 4, no. 3, pp. 602–610, Jul. 2013.
- [33] A. Rabiee, S. M. Mohseni-Bonab, T. Soltani, and L. Bayat, "A riskbased two-stage stochastic optimal power flow considering the impact of multiple operational uncertainties," *J. Energy Manage. Technol.*, vol. 1, no. 1, pp. 30–42, Jun. 2017.
- [34] M. A. Ortega-Vazquez and D. S. Kirschen, "Estimating the spinning reserve requirements in systems with significant wind power generation penetration," *IEEE Trans. Power Syst.*, vol. 24, no. 1, pp. 114–124, Feb. 2009.
- [35] W. Ko, D. Hur, and J.-K. Park, "Correction of wind power forecasting by considering wind speed forecast error," *J. Int. Council Electr. Eng.*, vol. 5, no. 1, pp. 47–50, Sep. 2015.
- [36] B. Mauch, J. Apt, P. M. S. Carvalho, and M. J. Small, "An effective method for modeling wind power forecast uncertainty," *Energy Syst.*, vol. 4, no. 4, pp. 393–417, Jun. 2013.
- [37] K. Hoyland and S. W. Wallace, "Generating scenario trees for multistage problems," *Manage. Sci.*, vol. 47, no. 2, pp. 295–307, Feb. 2001.
- [38] General Algebraic Modeling System (GAMS), Develop. Corp., Washington, DC, USA, 2013.
- [39] X. Bai, H. Wei, K. Fujisawa, and Y. Wang, "Semidefinite programming for optimal power flow problems," *Int. J. Electr. Power Energy Syst.*, vol. 30, nos. 6–7, pp. 383–392, Jul. 2008.

- [40] R. A. Jabr, "Radial distribution load flow using conic programming," *IEEE Trans. Power Syst.*, vol. 21, no. 3, pp. 1458–1459, Aug. 2006.
- [41] N. C. Koutsoukis, P. S. Georgilakis, and N. D. Hatziargyriou, "Service restoration of active distribution systems with increasing penetration of renewable distributed generation," *IET Gener, Transmiss. Distrib.*, vol. 13, no. 14, pp. 3177–3187, Jul. 2019.
- [42] C. Coffrin, H. L. Hijazi, and P. Van Hentenryck, "The QC relaxation: A theoretical and computational study on optimal power flow," *IEEE Trans. Power Syst.*, vol. 31, no. 4, pp. 3008–3018, Jul. 2016.
- [43] EDSO for Smart Grids. (Feb. 2018). Flexibility in the Energy Transition— A Toolbox for Electricity DSOs. Accessed: Nov. 20, 2019.[Online]. Available: https://www.edsoforsmartgrids.eu/flexibility-in-the-energytransition-a-toolbox-for-electricity-dsos/
- [44] Y. V. Makarov, C. Loutan, J. Ma, and P. de Mello, "Operational impacts of wind generation on California power systems," *IEEE Trans. Power Syst.*, vol. 24, no. 2, pp. 1039–1050, May 2009.
- [45] A. G. Tsikalakis, Y. A. Katsigiannis, P. S. Georgilakis, and N. D. Hatziargyriou, "Determining and exploiting the distribution function of wind power forecasting error for the economic operation of autonomous power systems," in *Proc. IEEE Power Eng. Soc. Gen. Meeting*, Montreal, QC, Canada, Jun. 2006, p. 8.
- [46] M. E. Baran and F. F. Wu, "Network reconfiguration in distribution systems for loss reduction and load balancing," *IEEE Trans. Power Del.*, vol. 4, no. 2, pp. 1401–1407, Apr. 1989.



VASILEIOS A. EVANGELOPOULOS received the Diploma degree in electrical and computer engineering from the National Technical University of Athens (NTUA), Greece, in 2013, and the M.Sc. degree in engineering-economic systems from NTUA, in 2016, where he is currently pursuing the Ph.D. degree with the School of Electrical and Computer Engineering. His research interests include power systems optimization, distributed energy resources, and flexibility in distribution

system operation. He is a member of the Technical Chamber of Greece.



IASON I. AVRAMIDIS (Student Member, IEEE) received the Diploma degree in electrical and computer engineering from the National Technical University of Athens, Greece, in 2017, and the M.Sc. degree in energy science and technology from ETH Zurich, Switzerland, in 2019. He is currently pursuing the Ph.D. degree with the Electrical Energy & Computer Architecture (ELECTA) Group, KU Leuven, Belgium. His research interests include planning and operation of smart dis-

tribution grids, and power systems optimization. He is a member of the Technical Chamber of Greece.



PAVLOS S. GEORGILAKIS (Senior Member, IEEE) received the Diploma and Ph.D. degrees in electrical and computer engineering from the National Technical University of Athens (NTUA), Athens, Greece, in 1990 and 2000, respectively. In September 2009, he joined as a Faculty Member of the School of Electrical and Computer Engineering, NTUA, where he has been an Associate Professor, since August 2018. From 2004 to 2009, he was an Assistant Professor with the School of

Production Engineering and Management, Technical University of Crete, Greece. From 1994 to 2003, he was with Schneider Electric AE, the Greek subsidiary of Schneider Electric, where he worked as the Quality Control Engineer, Transformer Design Engineer, Research and Development Manager, and Low Voltage Products Marketing Manager. His current research interests include optimization algorithms and computational intelligence techniques for the optimal operation and planning of smart distribution systems. He is a member of the Technical Chamber of Greece.